Author's response to reviews

Title: Biological Monitoring of Blood Naphthalene Level as a Marker of Occupational and Environmental Exposure to PAHs among Car-Mechanics and Car-Spray Painters in Rawalpindi

Authors:

Atif Kamal (dr.atifkamal@yahoo.com)
Mazhar Qayyum (dr.mazhar@yahoo.com)
Iqbal U Cheema (iucheema1962@yahoo.com)
Audil Rashid (audil@uaar.edu.pk)

Version: 6 Date: 30 May 2011

Author's response to reviews:

In response to Associate Editor’s Comment, following typographic and grammatical changes are made in the manuscript. The are highlighted in green.

1. At two places, "statistically not significant" is changed to "not statistically significant"
2. The abbreviation for “ppm”, is correctly written as “parts per million”.